Author's response to reviews

Title: Understanding "revolving door" patients in general practice: a qualitative study

Authors:

Andrea E Williamson Dr (andrea.williamson@glasgow.ac.uk)
Kenneth Mullen Dr (kenneth.mullen@glasgow.ac.uk)
Philip Wilson Dr (p.wilson@abdn.ac.uk)

Version: 2 Date: 22 January 2014

Author's response to reviews:

Please see attached letter